Author’s response to reviews

Title: C1GALT1 expression is associated with galactosylation of IgA1 in peripheral B lymphocyte in immunoglobulin A nephropathy

Authors:
Yue Xing (xingyue_51@163.com)
Lina Li (Lilina@medmail.com.cn)
Yaru Zhang (15122459564@163.com)
Fanghao Wang (wfh-92-05-09@163.com)
Dandan He (13132210716@163.com)
Youxia Liu (5liuyouxia@163.com)
Junya Jia (jiajunya@126.com)
Tiekun Yan (tjtiekun@126.com)
Shan Lin (linshan@medmail.com.cn)

Version: 1 Date: 10 Oct 2019

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to revise our manuscript entitled “C1GALT1 expression is associated with galactosylation of IgA1 in peripheral B lymphocyte in immunoglobulin A nephropathy (BNEP-D-19-00633)”. We appreciate all of the comments on this manuscript.

We have revised the manuscript according to the comments and suggestions. The responses to each comment are listed point-by-point below.

Editor Comments:

1. It is noticed that authors Dandan He, Junya Jia, Tiekun Yan, Shan Lin are not mentioned in the Authors’ contributions subsection of the Declarations. In this subsection please ensure all authors are referred to alongside their individual contributions to the study/manuscript.

Response: Thanks very much for your comments. We added the description of other authors’ contributions, including Dandan He, Junya Jia, and Tiekun Yan.
2. Currently your manuscript is in landscape format. Please revise and reformat so that your manuscript is in the standard Letter / A4 page format.

Response: Thanks very much. We revised our manuscript in the standard Letter / A4 page format.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Thanks very much. We uploaded our manuscript and all relevant tables/figures/additional files as a single, final, clean version.

Yours sincerely

ent of Nephrology, General Hospital of Tianjin Medical University
NO. 154, Anshan road, Heping District, Tianjin, China
Email: 5liuyouxia@163.com
Telephone: 86-22-60362531
Fax: 86-22-60362531